Author's response to reviews

Title: Medical issues in antenatal and perinatal periods among pregnant adolescents and young adults in Taiwan: a 10-year nationwide study

Authors:

Chun-Che Huang (huangmtc@gmail.com)
Yung-Chieh Lin (appledr@hotmail.com)
Yu-Tung Huang (yah@kmu.edu.tw)
Kuang-Hua Huang (khhuang@mail.cmu.edu.tw)

Version: 2 Date: 15 April 2014

Author's response to reviews: see over
April 15, 2014

Editor BMC Pregnancy and Childbirth

MS ID: 8515452981269917

Manuscript title: Medical issues in antenatal and perinatal periods among pregnant adolescents and young adults in Taiwan: a 10-year nationwide study

Authors: Huang et al.

Dear Editors,

Thank you for your letter dated April 15, 2014. We were pleased to know that our manuscript was rated as potentially acceptable for publication in *BMC Pregnancy and Childbirth*, subject to adequate revision and response to the comments raised by the editors.

As you notice, we have revised the manuscript by modifying the Line Numbering and Title Page sections, based on the comments made by the editors. Accordingly, we have uploaded a copy of the original manuscript marked with all the changes made during the revision process. The new text is revised and inserted line and page numbers.

Appended to this letter is our point-by-point response to the comments raised by the editors. As you notice, we agreed with all the comments raised by the editors. We would like also to thank you for allowing us to resubmit a revised copy of the manuscript.

I look forward to hearing from you at your earliest convenience.

Sincerely,

Yu-Tung Huang PhD
Master Degree Program in Aging and Long-Term Care,
Kaohsiung Medical University,
Kaohsiung, Taiwan
Telephone: 886-7-3121101 ext.2636
E-mail: yah@kmu.edu.tw
Editor Comments:

1. Please revise your manuscript to include line and page numbers.

[Reply]: As suggested, we corrected and inserted this into the text.

1. Please include all co-author's email addresses.

[Reply]: As suggested, we added on p. 1 of the revised manuscript.